UC Davis UC Davis Previously Published Works

Title

Dengue illness impacts daily human mobility patterns in Iquitos, Peru

Permalink https://escholarship.org/uc/item/2kk1v2d0

Journal PLOS Neglected Tropical Diseases, 13(9)

ISSN 1935-2727

Authors

Schaber, Kathryn L Paz-Soldan, Valerie A Morrison, Amy C <u>et al.</u>

Publication Date

2019

DOI

10.1371/journal.pntd.0007756

Peer reviewed



G OPEN ACCESS

Citation: Schaber KL, Paz-Soldan VA, Morrison AC, Elson WHD, Rothman AL, Mores CN, et al. (2019) Dengue illness impacts daily human mobility patterns in Iquitos, Peru. PLoS Negl Trop Dis 13(9): e0007756. https://doi.org/10.1371/journal. pntd.0007756

Editor: Roberto Barrera, Centers for Disease Control and Prevention, Puerto Rico, UNITED STATES

Received: May 1, 2019

Accepted: September 5, 2019

Published: September 23, 2019

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> <u>Commons CC0</u> public domain dedication.

Data Availability Statement: All data used in this study are available on Github (https://github.com/ KathrynSchaber/Movement-Data-Analysis) with the exception of study participant name, age, interview data, and spatial coordinates of home locations and locations visited, owing to the personal nature of that information. Code for fully reproducing the analyses in this study is also available on Github. RESEARCH ARTICLE

Dengue illness impacts daily human mobility patterns in Iquitos, Peru

Kathryn L. Schaber¹, Valerie A. Paz-Soldan², Amy C. Morrison³, William H. D. Elson³, Alan L. Rothman⁴, Christopher N. Mores⁵, Helvio Astete-Vega⁵, Thomas W. Scott³, Lance A. Waller⁶, Uriel Kitron⁷, John P. Elder⁸, Christopher M. Barker⁹, T. Alex Perkins¹⁰, Gonzalo M. Vazquez-Prokopec^{1,7}*

Program of Population Biology, Ecology and Evolution, Emory University, Atlanta, Georgia, United States of America, 2 Department of Global Community Health and Behavioral Sciences, Tulane School of Public Health and Tropical Medicine, New Orleans, Louisiana, United States of America, 3 Department of Entomology and Nematology, University of California Davis, Davis, California, United States of America,
Institute for Immunology and Informatics and Department of Cell and Molecular Biology, University of Rhode Island, Providence, Rhode Island, United States of America, 5 Department of Virology and Emerging Infections, U.S. Naval Medical Research Unit No. 6, Lima and Iquitos, Peru, 6 Department of Biostatistics and Bioinformatics, Rollins School of Public Health, Emory University, Atlanta, Georgia, United States of America,
Graduate School of Public Health, San Diego State University, San Diego, California, United States of America,
Graduate School of Public Health, University of Immunology, School of Veterinary Medicine,
University of California, Davis, California, United States of America, 10 Department of Biological Sciences and Eck Institute for Global Health, University of Notre Dame, Notre Dame, Indiana, United States of America

* gmvazqu@emory.edu

Abstract

Background

Human mobility plays a central role in shaping pathogen transmission by generating spatial and/or individual variability in potential pathogen-transmitting contacts. Recent research has shown that symptomatic infection can influence human mobility and pathogen transmission dynamics. Better understanding the complex relationship between symptom severity, infectiousness, and human mobility requires quantification of movement patterns throughout infectiousness. For dengue virus (DENV), human infectiousness peaks 0–2 days after symptom onset, making it paramount to understand human movement patterns from the beginning of illness.

Methodology and principal findings

Through community-based febrile surveillance and RT-PCR assays, we identified a cohort of DENV+ residents of the city of Iquitos, Peru (n = 63). Using retrospective interviews, we measured the movements of these individuals when healthy and during each day of symptomatic illness. The most dramatic changes in mobility occurred during the first three days after symptom onset; individuals visited significantly fewer locations (Wilcoxon test, p = 0.017) and spent significantly more time at home (Wilcoxon test, p = 0.005), compared to when healthy. By 7–9 days after symptom onset, mobility measures had returned to healthy levels. Throughout an individual's symptomatic period, the day of illness and their subjective

Funding: This research was supported by funding from the US National Institutes of Health-National Institute of Allergy and Infectious Diseases (NIH-NIAID) award number P01AI098670 (to T.W.S.). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

sense of well-being were the most significant predictors for the number of locations and houses they visited.

Conclusions/Significance

Our study is one of the first to collect and analyze human mobility data at a daily scale during symptomatic infection. Accounting for the observed changes in human mobility throughout illness will improve understanding of the impact of disease on DENV transmission dynamics and the interpretation of public health-based surveillance data.

Author summary

Dengue is the most important mosquito-borne viral disease of humans worldwide. Due to the limited mobility of the mosquitoes that transmit dengue virus, human mobility can be a key to both understanding an individual's exposure to the virus and explaining the spread of dengue throughout a population. Accurate disease models should include human mobility; however, changes in human movement patterns due to the presence of symptoms need to be taken into account. We quantified the impact of symptom presence on human mobility throughout the infectious period by analyzing a dataset on the daily movements of dengue virus infected individuals. Accounting for these changing patterns of mobility will improve understanding of the complex relationship between symptom severity, human movement, and dengue virus transmission. Furthermore, dengue transmission models that incorporate symptom-driven mobility changes can be used to evaluate scenarios and strategies for disease prevention.

Introduction

Human mobility plays a central role in shaping the structure of transmission networks and in influencing epidemiologic processes such as pathogen introduction, epidemic transmission, and endemic persistence [1-4]. While human mobility can drive transmission across multiple spatial and temporal scales [3, 5], it is at the finest scales (daily, intra-urban human movements) where epidemic processes occur and emergency public health interventions are usually implemented. Evidence from theoretical models and empirical studies show that individual and/or spatial variability in number and frequency of contacts can lead to transmission heterogeneity, where certain individuals or locations contribute disproportionately to pathogen transmission and epidemic spread [6-8]. Thus, identifying social and behavioral characteristics (e.g., mobility patterns, occupations, age classes) most responsible for such disproportionate contributions has become a public health priority, with significant potential for leveraging the power of public health surveillance programs and targeted disease control [9-11].

Dengue, an acute illness caused by four immunologically related viruses in the family *Flavi-viridae* and transmitted by *Aedes* spp. mosquitoes (primarily *Aedes aegypti*), is the most important mosquito-borne viral disease of humans worldwide [12]. Because *Aedes aegypti* seldom disperse beyond 100 meters, have a propensity for resting and biting inside residential buildings, and bite during the day [13–16], human movements are key to explaining the urban transmission dynamics of dengue virus (DENV) [17–21]. Individual movement patterns can also expand the spatial scale of transmission and lead to significant heterogeneity in transmission patterns by connecting otherwise discrete subpopulations of mosquitoes [3, 22, 23]. Extensive movement studies performed in the upper Amazon city of Iquitos, Peru, have shown that while human mobility within a resource-poor urban center is highly unstructured (with only 38% of participants having regular mobility routines), the majority of locations visited are either residential or commercial, with most movements (81%) occurring within 1 km of an individual's home [24–28]. Moreover, an individual's risk of DENV infection significantly increased when he or she routinely visited the same residential locations as DENV-infected people, whereas the distance the individual lived from a DENV-infected case was not significant [26].

Such empirical characterizations of fine-scale human mobility patterns and risk of DENV infection have informed complex simulation models that explore the theoretical role of human movement on the spatial and temporal patterns of disease dynamics [18, 23, 29]. When mobility-driven contact structure is included in theoretical models, the effect on DENV epidemic transmission is dramatic. Overlapping movement patterns within social groups drive the finescale heterogeneity in DENV transmission rates; however, the presence of a mosquito vector can hide the effect of socially structured movements if only spatially aggregated infection dynamics are considered [23]. Such models do not take into account the fact that symptomatic infection may influence mobility, which in turn can influence onward virus transmission and the structure of transmission chains. Research on directly transmitted diseases has demonstrated disease-driven behavior changes [30] and the significant influence they can have on predictions of pathogen spread [31-34]. For DENV, mobility data have been captured for febrile symptomatic individuals and healthy individuals using either retrospective movement surveys [35] or GPS trackers [36]. Febrile DENV-infected individuals visited significantly fewer places, traveled shorter distances, and spent more time at home [35, 36]. These patterns reveal particularly important information for understanding the complex relationship between symptom severity and human mobility, and to ultimately determine if there is an association between human mobility and infectiousness to mosquitoes. However, because DENV infectiousness peaks at 0-2 days after onset of symptoms and lasts for 4-5 days after onset of symptoms [37–39], human mobility during the first few days of symptoms could be key to better understand transmission dynamics. The goal of this study, therefore, was to conduct detailed, daily retrospective interviews to measure the mobility behavior of clinically apparent DENVinfected individuals throughout their illness, with the goal of generating mobility metrics that can be used to characterize the absolute and relative impacts of disease on potential exposure to Aedes aegypti mosquitoes.

Methods

Study area

This study was performed in the Amazon city of Iquitos, Peru. Iquitos is a geographically isolated, tropical urban environment with approximately 430,000 inhabitants located along the margin of the Amazon River [40]. The city's economic structure is highly informal and dynamic, with one-third of economically active individuals either unemployed or informally employed [41]. Iquitos has been the home of extensive, long-term arboviral research led by the University of California, Davis and U.S. Naval Medical Research Unit 6 since 1999 [3, 24–28, 42]. Extensive human mobility studies paired with detailed epidemiological data have made Iquitos an informative site for understanding the dynamics of arbovirus transmission. All four serotypes of DENV have been introduced in Iquitos; however, at any particular time virus transmission is usually dominated by a single serotype [42, 43]. Previous research [27] demonstrated that inhabitants visit an average of 5.8 (±3.6 SD) locations over a two-week period. While most movement (~80%) occurs within 1 km of their home, inhabitants have highly irregular and temporally unstructured routines that are not dominated by a single location, such as a workplace [27].

Study design

The study followed a contact-cluster design in which reverse transcription polymerase chain reaction (RT-PCR) positive, or viral nucleic acid test positive, DENV-infected individuals (index cases) were captured through community or clinic-based febrile surveillance systems, as described previously [26]. At the time of the initial blood sample, a 15-day retrospective semi-structured movement survey (RMS) was administered to the index case to identify the locations they visited in the 15 days prior to diagnosis (characterizing the "pre-illness" period). Consenting individuals (contacts) from the index cases' home and residential locations visited by the index case were then screened for DENV infection using RT-PCR [26] [44]. The RMS was administered to DENV PCR-positive contacts to quantify mobility behavior associated with potential virus exposure.

RMSs were developed based on findings of focus groups and validated by comparison with data from people wearing GPS tracking devices [24, 25]. RMSs capture positional, temporal and behavioral information of routine human mobility. Questions focus on the amount of time an individual spent at home, the visitors they received, and the places they visited. For time spent at home, individuals were asked about the average number of hours spent at home each day of the week, specifically focusing on the period from 5 a.m. to 10 p.m., which includes the peak landing and biting times for *Aedes aegypti* [45]. For places visited, information was collected on the type of place visited, when, for how long, and how often in the 15-day period. Trained, local Iquitos residents (the 'Movement Team') verbally administered electronic RMSs and recorded the information on tablets in the CommCare application [46].

To track movements of DENV positive participants during their illness, daily interviews using a modified daily RMS (DRMS) were conducted in person or by telephone for 7 days following the initial RT-PCR-positive blood test (S1 File). Where participants were not available for daily interviews, information about movements on several days was collected at a single interview. The DRMS asks about the amount of time spent at home the previous day(s) and the following information about each place visited during the previous day(s): day visited, place type, location, time of day visited, and time spent. For residential places visited, the DRMS asks whom they were visiting, their reason for visiting, if anyone in the home was ill during the preceding 15 days, and (for routinely visited houses) if/why there was any change in the time of visitation, as compared to the "pre-illness" period. During this seven-day period, DENV positive individuals were also administered two Quality of Well-Being surveys (QWB) by the Movement Team, one 2-3 days and one 7 days after the initial PCR-positive blood test. The QWB survey is a validated instrument used to measure an individual's quality of life during chronic illness [47]. Our study was a novel application of the QWB survey to an acute illness. The survey responses were sent to the developers at University of California, San Diego, who used a weighted algorithm to produce one well-being score between 0.0 (death) and 1.0 (asymptomatic and fully-functioning) covering the three days prior to each survey date [47].

At a follow-up visit scheduled 30 days after the initial PCR-positive blood test, individuals were given a 15-day ("post-illness") RMS and QWB survey in an effort to record their "base-line" mobility behavior and well-being in the absence of illness.

Data processing

For each study participant, the following variables were computed from the "pre-illness" and "post-illness" 15-day RMS: (1) total number of locations visited, (2) proportion of visits to

each location type, (3) total number of houses visited, (4) proportion of visits to houses of family members vs. houses of friends, and (5) average proportion of time spent at home per day. Equivalent daily values of these variables were collected for each participant from the DRMSs. Rather than referring to values as occurring on a certain number of days after the PCR-positive blood test, a standardized "day after symptom onset" variable was calculated. Because blood tests were not done on the same day of illness for all participants, DRMSs captured a range of 1–15 days after symptom onset. We focused our analysis on days 1–9 after symptom onset; few individuals had data for days 10–15 after symptom onset.

Data analysis

Analysis of mobility data had two main objectives: (1) comparing healthy (pre- and post- illness) mobility to mobility during illness, and (2) determining if mobility patterns changed during the 9 days after symptom onset.

For the first objective, mobility during illness was calculated by averaging a participant's DRMS for all available time points up to day 9 after symptom onset. Comparisons were done for the following mobility metrics: daily number of locations visited, daily number of houses visited, and proportion of time spent at home. When a metric followed a normal distribution (assessed via the Shapiro-Wilk test), pairwise comparisons were performed with paired t-tests followed by Holm-Bonferroni corrections. When the variable was not normally distributed, the non-parametric Kruskall-Wallis Rank Sum Test and pairwise Wilcoxon Signed Rank Test for paired data were utilized. As many individuals would stop visiting other locations during their illness period [35], we also analyzed the number of locations, number of houses, and time at home as binary variables, asking if any locations/houses were visited and if any time was spent away from home. These binary outcomes were compared between all possible pairs of time points (pre-during, during-post, and pre-post illness) using McNemar's χ^2 test.

If locations were visited, further analyses determined what type of locations they were. While these data were subject-correlated across time points, they could not be analyzed as paired data because not all participants visited locations at every time point. Generalized logistic mixed-effects models (GLMMs) determined the association between the probability of a location type being visited and the time period being considered (pre-during-post illness), while accounting for repeated measures by using participant ID as the random intercept. Location type was separated into four groups: (1) house, (2) health, (3) education/work, and (4) other (e.g., recreation, church, market, port). Similarly, logistic GLMMs determined the association between time period (pre-during-post symptoms) and the probability of a specific house type being visited (e.g., family versus friend).

For the second objective, aiming to determine whether mobility patterns changed during the illness period, we calculated mobility metrics for 3-day groups (days 1–3, 4–6, and 7–9 after onset of symptoms). Daily data were aggregated into 3-day groups to allow for robust analyses, while also controlling for the dearth of data points on certain days. In particular, the first two days after symptom onset had incomplete information for some participants due to the time required to capture individuals with symptoms, run RT-PCR tests, and obtain confirmed test results. To make pre/post and during-illness data comparable, 15-day RMS values were condensed to give movements over an average 3-day period. Analysis of the number of locations/houses, proportion of location/house types, and time spent at home followed the same steps described above. Comparisons were made between the 3-day groups to determine whether significant changes occur in movement patterns during illness. Further, movements within the 3-day groups were each compared to post-illness mobility.

The associations between daily (DRMS) mobility patterns and possible predictor variables were examined using Generalized Linear Mixed Models (GLMMs), Generalized Additive Mixed Models (GAMMs), and Generalized Additive Models for Location Scale and Shape (GAMLSSs) [48–50]. Best-fit models were determined for each of the following mobility outcomes: total number of locations visited (count variable), relative number of locations visited (compared to when healthy) (continuous variable), total number of houses visited (count), relative number of houses visited (continuous), total proportion of time spent time at home (continuous), and relative amount of time spent at home (continuous). For both the total number of locations visited and total number of houses visited, GLMMs and GAMMs with underlying Poisson distributions were compared. An individual's age, occupation, gender, QWB score, and the "day after symptom onset" were considered as predictor variables, with the best-fit model determined using a AIC and a Chi-square test comparing reduction in residual deviance. The response variable proportion of time spent at home was best characterized by a one-inflated beta distribution, so analysis was done with GAMLSS, as detailed below.

Although GLMM and GAMM regressions model the mean (μ) value of the distribution of the response variable, GAMLSS allows other distribution parameters to be modeled as a function of explanatory variables. A one-inflated beta distribution has possible values $0 < y \le 1$ and is defined in two parts: the probability that y = 1 (modeled by the η parameter) and the probability for 0 < y < 1, which is shaped by a traditional beta distribution with parameters mean (μ) and shape (σ). Here, the η parameter was the probability an individual stayed at home 100% of the time (y = 1). If an individual did not stay at home the entire day, the proportion of time that was spent at home (0 < y < 1) was determined by a beta distribution with μ and σ .

We also considered response variables as relative values in order to control for the individual variation in mobility levels. The number of locations (houses) an individual visited on each day during illness was considered relative to the average number of locations (houses) they visited pre-illness. Similarly, the number of hours a participant spent at home during each day of illness was compared to the average number of hours that individual spent at home pre-illness. While relative number of houses could not be well explained by a set distribution, both relative number of locations visited and relative amount of time spent at home were best characterized by the logistic distribution. Analysis of these response variables was done with GAMLSS, where both the mean (μ) and the standard deviation (σ) parameters of the logistic distribution could be modeled as a function of explanatory variables. Best-fit GAMLSS models were chosen using forward and backward selection for each of the explanatory variables. All statistical analyses were performed in R 3.3.0 statistical computing software [48–51].

Ethics statement

The procedures for enrollment of participants, dengue diagnosis, semi-structured interviews, and participant follow-up were approved by the Institutional Review Board (IRB) of the United States Naval Medical Research Center Unit No. 6 (NAMRU-6) (NAMRU6.2014.0021) in compliance with all applicable federal regulations governing the protection of human subjects. IRB relying agreements were established between NAMRU-6 and Emory University, Tulane University, University of California Davis, University of Rhode Island, San Diego State University, and University of Notre Dame. In addition to IRB approval, investigators obtained host country approval from the Loreto Regional Health Department, which oversees health research in Iquitos. Adult study participants provided written informed consent and a parent or guardian provided informed consent on behalf of child study participants.

Results

Detailed mobility data were collected from a total of 62 DENV+ participants. Descriptions of participant demographics and data completeness appear in the Supplemental Text. The most commonly reported symptoms were general malaise (100%), weakness (96.61%), fever (93.22%), headache (91.53%), anorexia (89.83%), and musculoskeletal pain (84.75%). During days 1–9 post-onset of symptoms, the average maximum malaise intensity of participants was 7.5 on a scale of 10 (range: 0–10), as compared to a mean intensity of 0.28 out of 10 (range: 0–6) during the post-illness time period. While all participants reported some level of malaise and dengue-related symptoms, the vast majority of participants (88.7%) received only outpatient care.

For the 34 participants with QWB scores collected at all time points (2–3 days post blood test, a week post blood test, and post-illness), scores were considered in terms of "days after symptom onset". The mean QWB score of those reported in the nine days after symptom onset was 0.61 (range: 0.25–1.0). The median QWB score was less than 0.70 all nine days after symptom onset; however, a few individuals had scores of 1.0 ("asymptomatic") as early as day 4. While the DRMS may have captured mobility on days when individuals were asymptomatic, the vast majority of individuals retained symptoms throughout the nine days after symptom onset (S3 Fig). Therefore, that period will be referred to as "during illness" in the remaining sections.

The mean QWB score at the post-illness time point was 0.88 (range: 0.48–1.0, median 1.0). While the post-illness survey may contain data from a few individuals who were still symptomatic, in general we considered it a suitable approximation of healthy mobility. Another proxy for healthy mobility, the "pre-illness" RMS, may have also captured some symptomatic movements because the 15-day retrospective survey could not be given until DENV cases were captured a median of 3 days after symptom onset.

Healthy vs. symptomatic mobility patterns

When comparing healthy (pre-/post-illness) and symptomatic time points, there was a significant difference in both the proportion of time spent at home and the average number of locations visited (Fig 1A and 1B). Healthy participants spent 60% of their time at home and visited an average of 1.3/1.1 (pre-/post-illness) locations per day, whereas ill participants spent 74% of their time at home (Wilcoxon test: p < 0.001) and visited an average of 0.73 locations (Wilcoxon test: pre-illness: p = 0.010) (S2 Table). Participants were also significantly less likely to visit other houses during illness, as compared to pre-illness (McNemar's χ^2 : p < 0.001) and post-illness (McNemar's χ^2 : p = 0.043) (S3 Table).

The odds (adjusted odds ratio/AOR) of an individual visiting an education/work location during healthy time points (AOR pre-/post-illness: 2.0/4.4) were significantly greater than during illness (GLMM: p < 0.001; Table 1, Fig 2). Similar significant differences were seen for visits to "other" place types (GLMM: p<0.001; Table 1, Fig 2). Conversely, the odds of participants going to a health-related place pre- or post-illness were significantly lower than during illness (AOR: pre-/post-illness: 0.019/0.002; Table 1, Fig 2). Although individuals were more likely to visit a house during the pre-illness time period as compared to during illness (AOR: 1.684; GLMM: p = 0.013), there was no significant difference for post-illness (AOR: 0.872; GLMM: p = 0.64), where individuals were predicted to visit houses with a mean probability of 21% (Table 1, S4 Table).

Mobility patterns during illness

During days 1–3 and 4–6 after symptom onset, individuals were significantly more likely to spend all of their time at home, compared to both days 7–9 after symptom onset (McNemar's



Fig 1. Mobility values throughout illness. (A) Average number of locations visited during each time period (B) Average proportion of time spent at home during each time period. Significant differences, denoted by letters, were detected using pairwise paired Wilcoxon Sign Rank tests with Bonferroni's correction to account for a family-wise error-rate of 0.05. All significant differences had p-values < 0.01.

https://doi.org/10.1371/journal.pntd.0007756.g001

 χ^2 : p = 0.046) and post-illness (McNemar's χ^2 : days 1–3: p = 0.008; days 4–6: p = 0.008) (S6 Table). There was also a significant difference in the average proportion of time spent for days 1–3 and 4–6 (76%) when compared to both days 7–9 (69%) (Wilcoxon test: days 1–3: p = 0.014; days 4–6: p = 0.008) and post-illness (59%) (Wilcoxon test: days 1–3: p = 0.005; days 4–6: p < 0.001; Fig 3B, S5 Table). Individuals were significantly less likely to visit any locations during illness compared to post-illness (McNemar's χ^2 : days 1–3: p = 0.001; days 4–6: p < 0.001; days 7–9: p = 0.008; S6 Table). Accordingly, the average number of locations visited was significantly lower on days 1–3 (paired t-test: p = 0.017) and 4–6 after symptom onset (paired t-test: p < 0.001) when compared to the mean 3.4 places visited every 3 days at post-

Table 1. Adjusted odds ratios (and 95% Confidence Intervals) for location type throughout illness, derived from logistic GLMMs.

	Pre-Illness (Day 0)	Post-Illness (Day 30)	
Education/Work	1.884 (CI: 1.317–2.725) ***	4.429 (CI: 2.964–6.695) ***	
Health	0.019 (CI: 0.009–0.037) ***	0.002 (CI: 0.001–0.008) ***	
Other	2.144 (CI: 1.496–3.110) *** 1.660 (CI: 1.109–2.508) *		
House	1.684 (CI: 1.110–2.585) *	0.872 (CI: 0.539-1.419)	
Family's House	0.348 (CI: 0.078-1.360)	0.509 (CI: 0.125-1.855)	

Odds Ratios are for pre- and post-illness time periods, compared to the period during illness (daily interview period) based on logistic GLMMs. For houses, the odds ratios are given for family member's houses (vs. friend's houses). Significant associations between time period and location visitation are denoted with asterisks (* p<0.05, ** p<0.01, ***p<0.001).

https://doi.org/10.1371/journal.pntd.0007756.t001



Fig 2. Place types visited throughout illness. Expressed as the percent of locations being visited of each location type. Other location type includes: markets, restaurants, ports, churches, cemeteries, recreational places, internet cafes, and all else. The number of participants who visited places is listed above each time point.

https://doi.org/10.1371/journal.pntd.0007756.g002

illness (Fig 3A). The average number of locations visited on days 1–3 (1.5 places/3-days) was also significantly less than the average number of locations visited on days 7–9 after symptom onset (2.2 places/3-days) (Wilcoxon test: p = 0.047; S5 Table).

When considering the type of location visited (Fig 4), the three during-illness time points (days 1-3/4-6/7-9) were compared to the post-illness period. Post-illness, the participants were predicted to visit education/work places with a 48% probability, "other" places with a 32% probability, houses with a 20% probability, and health-related places with only a 0.2% probability (S7 Table). Compared to post-illness, the odds of an individual visiting an education/ work place were significantly lower for days 1–3 (AOR: 0.08), days 4–6 (AOR: 0.22), and days 7–9 after symptom onset (AOR: 0.26) (GLMM: p < 0.001; Table 2). Conversely, the odds of visiting a health-related place during illness were significantly higher compared to post-illness (AOR: days 1–3: 3826; days 4–6: 1041; days 7–9: 365), likely due to the very low probability of a health-related location being visited post-illness when healthy (GLMM: p < 0.001). The



Fig 3. Mobility values during illness (in 3-day intervals). (A) Average number of locations visited per 3-day period. (B) Average proportion of time spent at home per 3-day period. Significant differences, denoted by letters, were detected using pairwise paired Wilcoxon Sign Rank tests with Bonferroni's correction to account for a family-wise error-rate of 0.05. All significant differences had p-values < 0.05.

https://doi.org/10.1371/journal.pntd.0007756.g003

likelihood of visiting a house during illness was not significantly different than the likelihood post-illness (AOR: days 1–3: 1.08; days 4–6: 1.20; days 7–9: 1.64; GLMM: p > 0.05). There were also no significant correlations between the illness time point and the odds of visiting a family member's (versus friend's) house (Table 2, S7 Table).

Daily mobility patterns during symptomatic illness

The best-fitting model to describe the relative number of locations visited was a GAMLSS with a logistic distribution. The μ parameter (mean) was best explained by a positive effect of day after symptom onset (p < 0.001) and a random intercept for participants, which allowed the mean relative number of locations to vary by participant. The σ parameter (standard deviation) was best explained by QWB score (p < 0.001), day after symptom onset (p < 0.001), and an interaction between the two (p < 0.001) (Table 3). For relative amount of time spent at home, the best-fit model was a GAMLSS with underlying logistic distribution, where the μ parameter was best explained by a negative effect of day (p = 0.0011) and a random intercept for participants. The σ parameter was best explained by a negative effect of day (p = 0.0011) and a random intercept for participants. The σ parameter was best explained by a positive effect of day after symptom onset (p < 0.001) (Table 4).

With proportion of time spent at home as the response variable, the best-fit model explains the η parameter as a function of age (<18 or >18) (p = 0.005) and an interaction between age and day after symptom onset (p = 0.018). The μ parameter was explained by a random slope of participants over time and the σ parameter was explained by the QWB score and a smoothed function of the day of illness (Table 5). This suggests that whether an individual spent all (100%) of their time at home was dependent on both their age and the day of illness, whereas the proportion of time spent at home (when less than 100%) depended on the day of illness





https://doi.org/10.1371/journal.pntd.0007756.g004

(p = 0.005) and how they were feeling (QWB score) (p = 0.094; Table 5). While the day of illness did not have an overall effect on the mean proportion of time spent at home (when less than 100%), the random slope for participants suggests that day of illness had a varying effect across participants.

Discussion

We found that dengue illness affects almost all aspects of an individual's mobility behavior. During mild symptomatic illness, individuals visited significantly fewer locations and houses and spent significantly more time at home. Further, symptomatic participants visited education/work and "other" locations less often than when they were healthy and visited health locations more often. These results (1) are consistent with and expand prior evidence indicating that individuals with symptomatic illness move less than healthy individuals [31, 35, 52];

	Days 1-3	Days 4–6	Days 7–9
Education/Work	0.08 (CI: 0.03-0.18)***	0.22 (CI: 0.11-0.43)***	0.26 (CI: 0.14-0.45)***
Health	3826 (CI: 585-42793)***	1041 (CI: 205-9101)***	365 (CI: 72-3175)***
Other	0.51 (CI: 0.22-1.08)	0.29 (CI: 0.14-0.56)***	0.86 (CI: 0.47-1.52)
House	1.08 (CI: 0.39-2.80)	1.20 (CI: 0.52-2.73)	1.64 (CI: 0.76-3.42)
Family's House	21.15 (CI: 0.16-1.48x10 ²³)	1.48 (CI: 0.22-12.35)	0.97 (CI: 0.15-6.40)

Table 2. Adjusted odds ratios (and 95% Confidence Intervals) for location type during illness, derived from logistic GLMMs.

Odds ratios are for days 1–3, 4–6, and 7–9 after symptom onset, as compared to the post-illness time period, based on logistic GLMMs. For houses, the odds ratios are given for family member's houses (vs. friend's houses). Significant associations between time period and location visitation are denoted with asterisks (* p<0.05, ** p<0.01, ***p<0.001).

https://doi.org/10.1371/journal.pntd.0007756.t002

(2) refine estimates of the effects of mild symptomatic dengue illness on movement by quantifying changes before, during and after the symptomatic phase of infection; and (3) suggest the need to better account for disease-driven mobility behavior changes in DENV transmission models [31, 53].

Table 3. Fixed effects of a GAMLS	S predicting relative number of locations	visited, as compared to pre-illness
-----------------------------------	---	-------------------------------------

Parameter		Estimate	Std. Error	z value	p-value
μ	Intercept	-1.07	0.023	-46.79	< 0.001 ***
μ	Day	0.063	0.0132	4.81	< 0.001 ***
σ	Intercept	-7.001	0.914	14.85	< 0.001 ***
σ	Day	1.086	0.157	-2.85	0.001 ***
σ	QWB score	7.371	5.07	1.70	0.001 ***
σ	Day*QWB score	-1.252	-5.56	1.70	0.001 ***

Each distribution parameter (μ,σ) has distinct explanatory variables.

 μ , mean of logistic distribution; σ , standard deviation of logistic distribution

* p < 0.05,

** p < 0.01,

*** p < 0.001

https://doi.org/10.1371/journal.pntd.0007756.t003

Table 4. Fixed effects of a GAMLSS predicting relative amount of time spent at home, as compared to pre-illness.

Parameter		Estimate	Std. Error	z value	p-value
μ	Intercept	4.974	0.385	12.91	< 0.001 ***
μ	Day	-0.295	0.088	-3.37	0.0011 **
σ	Intercept	-0.133	0.164	-0.81	0.421
σ	Day	0.107	0.029	3.69	< 0.001 ***

Each distribution parameter (μ, σ) has distinct explanatory variables.

 μ , mean of logistic distribution; σ , standard deviation of logistic distribution

* p < 0.05,

** p < 0.01,

*** p < 0.001

https://doi.org/10.1371/journal.pntd.0007756.t004

Parameter		Estimate	Std. Error	z value	p-value
μ	Intercept	0.599	1.02e-05	58426	< 0.001 ***
σ	Intercept	9.147	0.616	14.85	< 0.001 ***
σ	pb(Day)	-0.332	0.116	-2.85	0.006 **
σ	QWB score	1.754	1.034	1.70	0.094
η	Intercept	-1.995	0.930	-2.15	0.035 *
η	Age: (<18)	2.992	1.044	2.87	0.005 **
η	Day	0.215	0.154	1.40	0.166
η	Age: (<18)*Day	-0.425	0.176	-2.42	0.018 *

Table 5. Fixed effects of a GAMLSS predicting proportion of time spent at home.

Each distribution parameter (μ,σ,η) has distinct explanatory variables.

 $\mu,$ mean of beta distribution; $\sigma,$ precision of beta distribution; $\eta,$ parameter for probability at one

** p < 0.01,

*** p < 0.001

https://doi.org/10.1371/journal.pntd.0007756.t005

The most dramatic changes in mobility occurred during the first 3 days after symptom onset, when significantly fewer locations were visited and significantly more time was spent at home. During days 4-6 and 7-9 after symptom onset, the number of locations visited increased and the proportion of time spent at home decreased. By days 7-9 after symptom onset, the number of locations visited and the time spent at home were no longer significantly different from healthy behaviors. This reduction in mobility during illness, particularly on days 1-3 after symptom onset, could affect an individual's contribution to onwards DENV transmission. For DENV, viremia reaches levels infectious to mosquitoes a few days prior to symptom onset and peaks at 0-2 days after symptom onset, with titers then lowering by days 4-5 (although some individuals are still capable of infecting mosquitoes) (37-39). During peak infectiousness, most individuals are spending more time at home and visiting fewer places, thereby reducing the number of distinct Aedes aegypti mosquitoes with whom potential virusspreading contacts occur. This may allow those few individuals who do not alter their movements to have a more significant role in pathogen transmission during peak infectiousness. During the pre-symptomatic period, however, almost all individuals have high mobility and a viremia level sufficient for virus transmission to mosquitoes [38]. Recent theoretical models of within-host viral dynamics for symptomatic individuals estimate that 24% of onward transmission results from mosquitoes biting during the pre-symptomatic period [54]. When also accounting for mobility changes throughout viremia, many individuals may have their greatest contribution to transmission be during the pre-symptomatic stage. Ten Bosch et. al. also estimated that asymptomatic individuals had only 80% the net infectiousness of symptomatic individuals [54]. This reduction in net infectiousness may be counteracted by the hypothetically unaltered mobility patterns exhibited by asymptomatic (and minimally symptomatic) individuals, further increasing the overall contribution of silent transmission. Such potential dynamics emerging from the coupling between individual infectiousness, movement, and disease severity deserve further investigation [55], because they may help explain the explosive nature of DENV outbreaks and the limitations of vector control in containing virus transmission.

Throughout an individual's illness period, we found that day of illness and the participant's subjective sense of well-being (QWB score) were significant predictors for the relative number of locations visited, as compared to pre-illness. When considering the proportion of time

^{*} p < 0.05,

spent at home, an individual's age and their day of illness were significant in predicting whether they chose to stay at home 100% of the day or not, with children being more likely to stay home all day compared to adults. When an individual chose to spend some amount of time outside their house, the day of illness and the QWB score significantly predicted the proportion of time. Further, the relative amount of time participants spent at home compared to pre-illness was also significantly predicted by the day of illness. Individuals with more severe symptoms and those at the beginning of their illness were more likely to be spending more time at home (both absolute proportion of time and amount of time relative to pre-illness values). Further, when compared to pre-illness, individuals at the beginning of their illness have lower values of relatively visited locations compared to toward the end of illness.

One limitation of our study is the reliance on participant recall, which can be subject to recall bias. However, the retrospective semi-structured interview we utilized was previously tested in Iquitos and was found to obtain superior data on activity space, as compared to wearable GPS data-loggers [24]. Further, in the DRMS participants only needed to recall movements over the past 24 hours, making bias less likely. Our study also faced limitations with the number of participants and the ability to measure movement on the first two days after symptom onset. Nevertheless, our study is one of the first to collect human mobility data at a daily scale during symptomatic infection. Future studies could build on our study by collecting detailed mobility data from more individuals with a wider spectrum of symptom severity, including across a wider range of diseases. Future studies should also seek to make coupled measurements of an individual's infectiousness throughout the course of mobility data collection.

Human mobility patterns have played an important role in recent vector-borne disease transmission models [56]. There is, however, an increasing need to include differing mobility patterns when modeling individuals that are ill versus healthy. We demonstrate that individuals with dengue spend significantly more time at home, particularly during the first days after symptom onset when they are most infectious, potentially limiting contact with Ae. aegypti outside their home. When looking at the locations being visited during illness, however, the proportion of houses was consistent throughout and remained similar to the post-illness level. This may be of particular importance for onward transmission given the propensity for Ae. *aegypti* to bite inside houses [13, 14, 57]. The abundance of mosquitoes in both an individual's home and the houses/locations they visit when infectious will likely determine the effect that reduced mobility has on their overall contribution to DENV transmission. Reduction in mobility patterns when symptomatic could also affect the amount of overlap a social group has in the places they frequent. Given the significant role of socially structured human mobility in determining fine-scale DENV transmission rates [23], accounting for the dynamic nature of social contacts during a symptomatic DENV infection could allow for more accurate modeling of disease transmission and the design of more efficient disease prevention strategies.

Supporting information

S1 Text. Participant description. Gives participant demographics and data on the number of movement surveys collected at each time point. (DOCX)

S1 Table. Number of participants with data on each day post-symptom onset. In the bottom section, the number of participants for each 3-day group is given. (PDF)

S2 Table. Results of pairwise Wilcoxon Sign Rank tests of paired data for time points pre-, during, and post-illness. Tests were performed for number of locations visited, number of houses visited, and proportion of time spent at home, comparing between three time points: pre-, during, and post-illness. (* p<0.05, ** p<0.01, ***p<0.001). (PDF)

S3 Table. Results of McNemar's χ^2 test for time points pre-, during, and post-illness. Tests were performed for number of locations visited, number of houses visited, and proportion of time spent at home, comparing between three time points: pre-,during, and post-illness. (* p<0.05, ** p<0.01, ***p<0.001). (PDF)

S4 Table. Mean predicted probability of a specific location type being visited throughout illness. Probabilities are predicted for pre-, during, and post-illness time periods, based on logistic GLMMs. For houses, the probabilities are predicted for visiting family member's houses (vs friend's houses) (Table 1). (PDF)

S5 Table. Results of pairwise Wilcoxon Sign Rank tests of paired data for time points during illness. Tests were performed for number of locations visited, number of houses visited, and proportion of time spent at home, comparing between post-illness and three time points during illness (days 1–3, 4–6, 7–9). (* p<0.05, ** p<0.01, ***p<0.001). (PDF)

S6 Table. Results of McNemar's χ^2 test for time points during illness. Tests were performed for number of locations visited, number of houses visited, and proportion of time spent at home, comparing between four time points: days 1–3, 4–6, 7–9, and post-illness. (* p<0.05, ** p<0.01, ***p<0.001). (PDF)

S7 Table. Mean predicted probability of a specific location type being visited during illness. Probabilities are predicted for time points during illness and post-illness, based on logistic GLMMs. For houses, the probabilities are predicted for visiting family member's houses (vs friend's houses) (Table 2).

(PDF)

S8 Table. Results from likelihood ratio tests between pairs of GLMMs of total number of locations visited with various explanatory variables. The Chi square test statistic is looking at the reduction in deviance for each model as compared to GLMM(day). AICs are also provided for each model. The best-fit model is highlighted in red. (PDF)

S9 Table. Table comparing additive regression models for total number of locations visited with various explanatory variables. AIC values, degrees of freedom (DF), and amount of deviance explained (%) are provided for each model. The best-fit model is highlighted in red. (PDF)

S10 Table. Fixed effects of the best-fit model for total number of locations visited: GLMM "Total Locations ~ day". (PDF)

S11 Table. Results from likelihood ratio tests between pairs of GLMMs of total number of houses visited with various explanatory variables. The Chi square test statistic is looking at

the reduction in deviance for each model as compared to GLMM(day). AICs are also provided for each model. The best-fit model is highlighted in red. (PDF)

S12 Table. Table comparing additive regression models for total number of houses visited with various explanatory variables. AIC values, degrees of freedom (DF), and amount of deviance explained (%) are provided for each model. The best-fit model is highlighted in red. (PDF)

S13 Table. Fixed effects of the best-fit model for total number of houses visited: GLMM "Total Houses ~ day".

(PDF)

S1 Fig. Average number of houses visited throughout illness. Expressed as the average number of houses visited during each time period. Significant differences, denoted by letter, were found using pairwise paired t-tests with Holm's correction to account for a family-wise error-rate of 0.05.

(TIF)

S2 Fig. Average number of houses visited during illness (in 3-day intervals). Expressed as the average number of locations visited per 3-day period for time point. Significant differences, denoted by letter, were found using pairwise paired t-tests with Holm's correction to account for a family-wise error-rate of 0.05.

(TIFF)

S3 Fig. Median (interquartile range) QWB-score as a function of day of illness. Numbers on top indicate number of surveys that included paired movement and QWB data. (TIFF)

S1 File. Daily Retrospective Movement Survey (DRMS). Lists all survey questions and all possible answers for multiple-choice questions. (PDF)

S2 File. XML code for Daily Retrospective Movement Survey (DRMS). XML code to recreate DRMS survey.

(TXT)

S1 Checklist. STROBE Checklist. Checklist used for reporting observational studies. (DOC)

Acknowledgments

We thank the residents of Iquitos for their support and participation in the study. We thank Sarah Jenkins, Bob Hontz, our Peruvian movement team (Jhonny Jorge Cordova-López, Esther Jennifer Rios López, Wendy Lorena Quiroz Flores, Alfonso Simone Vizcarra Santillan), project physicians (Crystyan Alls Neufert Siles Montoya, Ali Yalta, Isabel Bazan), data management personnel (Gabriela Vasquez De La Torre, Magaly Ochoa Isuiza), and the nurse technicians involved in case capture (Llerme Armas Pisco, Cesar Augusto Banda Chavez, Linder Maria Canayo Zavaleta, Clara Chávez López, Karina Chuquipiondo Vasquez, Laury Dacia Cuespan Camus, Leny Curico Manihuari, Nadia Rocio Del Rio Chavez, Salome Elespuru, Junnelhy Mireya Flores López, Juan Flores Michi, Luz Angelica Galvez, Huayllahua, Rina Gonzales Jaba, Deisy Irene Huiñapi Cambunugue, Maria Edith Juárez Baldera, Xiomara Mafaldo García, Nora Marín Romero, Nadia Tereshkova Montes Criollo, Johnni Mozombite Flores, Sandra Ivonne Moñoz Perez, Lucy Navarro Sánchez, Geraldine Ocmin Galán, Zenith María Pezo Villacorta, Fiona Stefani Pinedo Zevallos, Iris Reátegui Carrión, Zoila Martha Reategui Chota, Sadith Jovita Ricopa Manuyama, Liliana Rios López, Rubiela Nerza Rubio Briceño, Ysabel Ruiz Berger, Miranda Angela Rocio Soplin, Rosana Magaly Sotero Jiménez, Rosa Tamani Babilonia, Zenith Tamani Guerrero, Moises Tanchiva, Sarita Del Pilar Tuesta Dávila, Flora Vargas Ceras, Rita Gabriela Vasquez Orbe).

Copyright statement

Author H.A.V. is a military service member (or employee of the U.S. Government). This work was prepared as part of his official duties. Title 17 U.S.C. §105 provides that 'Copyright protection under this title is not available for any work of the United States Government.' Title 17 U.S.C. §101 defines a U.S. Government work as a work prepared by a military service member or employee of the U.S. Government as part of that person's official duties.

Disclaimer

The views expressed in this article are those of the author and do not necessarily reflect the official policy or position of the Department of the Navy, Department of Defense, nor the U.S. Government.

Author Contributions

Conceptualization: Valerie A. Paz-Soldan, Alan L. Rothman, Thomas W. Scott, John P. Elder, T. Alex Perkins, Gonzalo M. Vazquez-Prokopec.

- **Data curation:** Kathryn L. Schaber, Amy C. Morrison, William H. D. Elson, Christopher N. Mores, Christopher M. Barker, Gonzalo M. Vazquez-Prokopec.
- Formal analysis: Kathryn L. Schaber.
- Funding acquisition: Valerie A. Paz-Soldan, Amy C. Morrison, Alan L. Rothman, Thomas W. Scott, Uriel Kitron, John P. Elder, Christopher M. Barker, Gonzalo M. Vazquez-Prokopec.
- Investigation: Valerie A. Paz-Soldan, Amy C. Morrison, William H. D. Elson, Helvio Astete-Vega.
- Methodology: Valerie A. Paz-Soldan, Amy C. Morrison, Alan L. Rothman, Christopher N. Mores, Thomas W. Scott, John P. Elder, T. Alex Perkins, Gonzalo M. Vazquez-Prokopec.
- Project administration: Valerie A. Paz-Soldan, Amy C. Morrison, Alan L. Rothman, Christopher N. Mores, Helvio Astete-Vega, Thomas W. Scott, John P. Elder, Gonzalo M. Vazquez-Prokopec.

Resources: Valerie A. Paz-Soldan, Amy C. Morrison, Alan L. Rothman, Christopher N. Mores, Thomas W. Scott, John P. Elder, Gonzalo M. Vazquez-Prokopec.

- Software: William H. D. Elson, Christopher M. Barker.
- Supervision: Valerie A. Paz-Soldan, Amy C. Morrison, William H. D. Elson, Alan L. Rothman, Helvio Astete-Vega, Thomas W. Scott, John P. Elder, Gonzalo M. Vazquez-Prokopec.

Validation: Amy C. Morrison, William H. D. Elson, Alan L. Rothman.

Visualization: Kathryn L. Schaber.

Writing - original draft: Kathryn L. Schaber, Gonzalo M. Vazquez-Prokopec.

Writing – review & editing: Kathryn L. Schaber, Valerie A. Paz-Soldan, Amy C. Morrison, William H. D. Elson, Alan L. Rothman, Christopher N. Mores, Helvio Astete-Vega, Thomas W. Scott, Lance A. Waller, Uriel Kitron, John P. Elder, Christopher M. Barker, T. Alex Perkins, Gonzalo M. Vazquez-Prokopec.

References

- Eubank S, Guclu H, Kumar VS, Marathe MV, Srinivasan A, Toroczkai Z, et al. Modelling disease outbreaks in realistic urban social networks. Nature. 2004; 429(6988):180–4. Epub 2004/05/14. https://doi. org/10.1038/nature02541 PMID: 15141212.
- González M, Hidalgo C, Barabási A. Understanding individual human mobility patterns. Nature. 2008; 453:779–82. https://doi.org/10.1038/nature06958 PMID: 18528393
- 3. Stoddard ST, Morrison AC, Vazquez-Prokopec GM, Paz Soldan V, Kochel TJ, Kitron U, et al. The role of human movement in the transmission of vector-borne pathogens. PLoS neglected tropical diseases. 2009; 3(7):e481. https://doi.org/10.1371/journal.pntd.0000481 PMID: 19621090
- Kucharski AJ, Kwok KO, Wei VW, Cowling BJ, Read JM, Lessler J, et al. The contribution of social behaviour to the transmission of influenza A in a human population. PLoS Pathog. 2014; 10(6): e1004206. https://doi.org/10.1371/journal.ppat.1004206 PMID: 24968312
- Riley S. Large-scale spatial-transmission models of infectious disease. Science. 2007; 316 (5829):1298–301. Epub 2007/06/02. https://doi.org/10.1126/science.1134695 PMID: 17540894.
- Bansal S, Grenfell BT, Meyers LA. When individual behaviour matters: homogeneous and network models in epidemiology. Journal of the Royal Society, Interface / the Royal Society. 2007; 4(16):879– 91. https://doi.org/10.1098/rsif.2007.1100 PMID: 17640863
- Arthur RF, Gurley ES, Salje H, Bloomfield LS, Jones JH. Contact structure, mobility, environmental impact and behaviour: the importance of social forces to infectious disease dynamics and disease ecology. Philos Trans R Soc Lond B Biol Sci. 2017; 372(1719). https://doi.org/10.1098/rstb.2016.0454 PMID: 28289265
- Keeling MJ, Danon L, Vernon MC, House TA. Individual identity and movement networks for disease metapopulations. Proceedings of the National Academy of Sciences of the United States of America. 2010; 107(19):8866–70. Epub 2010/04/28. https://doi.org/10.1073/pnas.1000416107 PMID: 20421468
- 9. Vespignani A. Predicting the behavior of techno-social systems. Science. 2009; 325(5939):425–8. Epub 2009/07/25. https://doi.org/10.1126/science.1171990 PMID: 19628859.
- Eames KT, Keeling M. Modeling Dynamic and Network Heterogeneities in the Spread of Sexually Transmitted Diseases. Proc Natl Acad Sci. 2002; 99(20):13330–5. <u>https://doi.org/10.1073/pnas.</u> 202244299 PMID: 12271127
- 11. Eames KT, Keeling MJ. Contact tracing and disease control. Proc Biol Sci. 2003; 270(1533):2565–71. https://doi.org/10.1098/rspb.2003.2554 PMID: 14728778
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496(7446):504–7. https://doi.org/10.1038/nature12060 PMID: 23563266
- Chadee DD, Gilles JR. The diel copulation periodicity of the mosquito, Aedes aegypti (L.) (Diptera: Culicidae) at indoor and outdoor sites in Trinidad, West Indies. Acta Trop. 2014; 132 Suppl:S91–5. <u>https://doi.org/10.1016/j.actatropica.2013.06.022</u> PMID: 23850504.
- 14. Chadee DD, Sutherland JM, Gilles JR. Diel sugar feeding and reproductive behaviours of Aedes aegypti mosquitoes in Trinidad: with implications for mass release of sterile mosquitoes. Acta Trop. 2014; 132 Suppl:S86–90. https://doi.org/10.1016/j.actatropica.2013.09.019 PMID: 24076041.
- Scott TW, Amerasinghe PH, Morrison AC, Lorenz LH, Clark GG, Strickman D, et al. Longitudinal studies of Aedes aegypti (Diptera: Culicidae) in Thailand and Puerto Rico: blood feeding frequency. J Med Entomol. 2000; 37(1):89–101. Epub 2004/06/29. https://doi.org/10.1603/0022-2585-37.1.89 PMID: 15218911.
- Harrington LC, Scott TW, Lerdthusnee K, Coleman RC, Costero A, Clark GG, et al. Dispersal of the dengue vector Aedes aegypti within and between rural communities. The American journal of tropical medicine and hygiene. 2005; 72(2):209–20. Epub 2005/03/03. PMID: <u>15741559</u>.
- Falcon-Lezama JA, Martinez-Vega RA, Kuri-Morales PA, Ramos-Castaneda J, Adams B. Day-to-Day Population Movement and the Management of Dengue Epidemics. Bulletin of mathematical biology. 2016; 78(10):2011–33. https://doi.org/10.1007/s11538-016-0209-6 PMID: 27704330
- Nevai AL, Soewono E. A model for the spatial transmission of dengue with daily movement between villages and a city. Math Med Biol. 2014; 31(2):150–78. https://doi.org/10.1093/imammb/dqt002 PMID: 23475426

- Perkins TA, Scott TW, Le Menach A, Smith DL. Heterogeneity, mixing, and the spatial scales of mosquito-borne pathogen transmission. PLoS computational biology. 2013; 9(12):e1003327. <u>https://doi.org/10.1371/journal.pcbi.1003327 PMID: 24348223</u>
- Kraemer MU, Perkins TA, Cummings DA, Zakar R, Hay SI, Smith DL, et al. Big city, small world: density, contact rates, and transmission of dengue across Pakistan. Journal of the Royal Society, Interface / the Royal Society. 2015; 12(111):20150468. https://doi.org/10.1098/rsif.2015.0468 PMID: 26468065
- Vazquez-Prokopec GM, Montgomery BL, Horne P, Clennon JA, Ritchie SA. Combining contact tracing with targeted indoor residual spraying significantly reduces dengue transmission. Science advances. 2017; 3(2):e1602024. Epub 2017/02/25. https://doi.org/10.1126/sciadv.1602024 PMID: 28232955
- Adams B, Kapan D. Man Bites Mosquito: Understanding the Contribution of Human Movement to Vector-Borne Disease Dynamics. PloS one. 2009; 4(8):1–10. https://doi.org/10.1371/journal.pone.0006763 PMID: 19707544
- Reiner RC Jr., Stoddard ST, Scott TW. Socially structured human movement shapes dengue transmission despite the diffusive effect of mosquito dispersal. Epidemics. 2014; 6:30–6. <u>https://doi.org/10.1016/j.epidem.2013.12.003</u> PMID: 24593919
- Paz Soldan V, Reiner RC Jr., Morrison AC, Stoddard ST, Kitron U, Scott TW, et al. Strengths and Weaknesses of Global Positioning System (GPS) Data-Loggers and Semi-structured Interviews for Capturing Fine-scale Human Mobility: Findings from Iquitos, Peru. PLoS neglected tropical diseases. 2014; 8(6):e2888. https://doi.org/10.1371/journal.pntd.0002888 PMID: 24922530
- 25. Paz-Soldan VA, Stoddard ST, Vazquez-Prokopec G, Morrison AC, Elder JP, Kitron U, et al. Assessing and maximizing the acceptability of global positioning system device use for studying the role of human movement in dengue virus transmission in lquitos, Peru. The American journal of tropical medicine and hygiene. 2010; 82(4):723–30. https://doi.org/10.4269/ajtmh.2010.09-0496 PMID: 20348526
- 26. Stoddard ST, Forshey BM, Morrison AC, Paz Soldan V, Vazquez-Prokopec GM, Astete H, et al. Houseto-house human movement drives dengue virus transmission. Proc Natl Acad Sci. 2013; 110(3):994–9. https://doi.org/10.1073/pnas.1213349110 PMID: 23277539
- Vazquez-Prokopec GM, Bisanzio D, Stoddard ST, Paz-Soldan V, Morrison AC, Elder JP, et al. Using GPS technology to quantify human mobility, dynamic contacts and infectious disease dynamics in a resource-poor urban environment. PloS one. 2013; 8(4):e58802. https://doi.org/10.1371/journal.pone. 0058802 PMID: 23577059
- Vazquez-Prokopec GM, Stoddard ST, Paz-Soldan V, Morrison AC, Elder JP, Kochel TJ, et al. Usefulness of commercially available GPS data-loggers for tracking human movement and exposure to dengue virus. Int J Health Geogr. 2009; 8:68. Epub 2009/12/02. <u>https://doi.org/10.1186/1476-072X-8-68</u> PMID: 19948034
- Wesolowski A, Qureshi T, Boni MF, Sundsoy PR, Johansson MA, Rasheed SB, et al. Impact of human mobility on the emergence of dengue epidemics in Pakistan. Proceedings of the National Academy of Sciences of the United States of America. 2015; 112(38):11887–92. <u>https://doi.org/10.1073/pnas.</u> 1504964112 PMID: 26351662
- Poletti P, Visintainer R, Lepri B, Merler S. The interplay between individual social behavior and clinical symptoms in small clustered groups. BMC infectious diseases. 2017; 17(1):521. <u>https://doi.org/10. 1186/s12879-017-2623-2 PMID: 28747154</u>
- Van Kerckhove K, Hens N, Edmunds WJ, Eames KT. The impact of illness on social networks: implications for transmission and control of influenza. American journal of epidemiology. 2013; 178(11):1655– 62. https://doi.org/10.1093/aje/kwt196 PMID: 24100954
- Wang Z, Andrews MA, Wu ZX, Wang L, Bauch CT. Coupled disease-behavior dynamics on complex networks: A review. Phys Life Rev. 2015; 15:1–29. <u>https://doi.org/10.1016/j.plrev.2015.07.006</u> PMID: 26211717.
- Poletto C, Tizzoni M, Colizza V. Human mobility and time spent at destination: impact on spatial epidemic spreading. J Theor Biol. 2013; 338:41–58. https://doi.org/10.1016/j.jtbi.2013.08.032 PMID: 24012488.
- Fenichel EP, Castillo-Chavez C, Ceddia MG, Chowell G, Gonzalez Parra PA, Hickling GJ, et al. Adaptive Human Behavior in Epidemiological Models. Proc Natl Acad Sci. 2011; 108(15):6306–11. <u>https:// doi.org/10.1073/pnas.1011250108</u> PMID: 21444809
- Perkins TA, Paz-Soldan VA, Stoddard ST, Morrison AC, Forshey BM, Long KC, et al. Calling in sick: impacts of fever on intra-urban human mobility. Proc Biol Sci. 2016; 283(1834). https://doi.org/10.1098/ rspb.2016.0390 PMID: 27412286
- Falcon-Lezama JA, Santos-Luna R, Roman-Perez S, Martinez-Vega RA, Herrera-Valdez MA, Kuri-Morales AF, et al. Analysis of spatial mobility in subjects from a Dengue endemic urban locality in Morelos State, Mexico. PloS one. 2017; 12(2):e0172313. <u>https://doi.org/10.1371/journal.pone.0172313</u> PMID: 28225820

- Clapham HE, Tricou V, Van Vinh Chau N, Simmons CP, Ferguson NM. Within-host viral dynamics of dengue serotype 1 infection. Journal of the Royal Society, Interface / the Royal Society. 2014; 11(96). https://doi.org/10.1098/rsif.2014.0094 PMID: 24829280
- Duong V, Lambrechts L, Paul RE, Ly S, Lay RS, Long KC, et al. Asymptomatic humans transmit dengue virus to mosquitoes. Proceedings of the National Academy of Sciences of the United States of America. 2015; 112(47):14688–93. Epub 2015/11/11. <u>https://doi.org/10.1073/pnas.1508114112</u> PMID: 26553981
- 39. Nguyet MN, Duong TH, Trung VT, Nguyen TH, Tran CN, Long VT, et al. Host and viral features of human dengue cases shape the population of infected and infectious Aedes aegypti mosquitoes. Proceedings of the National Academy of Sciences of the United States of America. 2013; 110(22):9072–7. Epub 2013/05/16. https://doi.org/10.1073/pnas.1303395110 PMID: 23674683
- **40.** Instituto Nacional de Estadística e Informática. Perú: Estimaciones y proyecciones de población total por sexo de las principales ciudades, 2012–2015. 2012.
- **41.** Instituto Nacional de Estadística e Informática. Censos Nacionales 2007: XI de Población y VI de Vivienda. Lima, Peru. 2008.
- Morrison AC, Minnick SL, Rocha C, Forshey BM, Stoddard ST, Getis A, et al. Epidemiology of dengue virus in Iquitos, Peru 1999 to 2005: interepidemic and epidemic patterns of transmission. PLoS neglected tropical diseases. 2010; 4(5):e670. https://doi.org/10.1371/journal.pntd.0000670 PMID: 20454609
- 43. Stoddard ST, Wearing HJ, Reiner RC Jr., Morrison AC, Astete H, Vilcarromero S, et al. Long-term and seasonal dynamics of dengue in lquitos, Peru. PLoS neglected tropical diseases. 2014; 8(7):e3003. https://doi.org/10.1371/journal.pntd.0003003 PMID: 25033412
- Forshey BM, Guevara C, Laguna-Torres VA, Cespedes M, Vargas J, Gianella A, et al. Arboviral etiologies of acute febrile illnesses in Western South America, 2000–2007. PLoS neglected tropical diseases. 2010; 4(8):e787. https://doi.org/10.1371/journal.pntd.0000787 PMID: 20706628
- Chadee DD, Martinez R. Landing periodicity of Aedes aegypti with implications for dengue transmission in Trinidad, West Indies. J Vector Ecol. 2000; 25(2):158–63. PMID: 11217215
- 46. CommCare by Dimagi | Data Collection App [Internet]. Dimagi. https://www.dimagi.com/commcare.
- 47. Seiber W, Groessl E, David K, Ganiats T, Kaplan R. Quality of Well-Being Self-Administered (QWB-SA) Scale: User's Manual. Health Services Research Center, University of California, San Diego, 2008.
- Rigby RA, Stasinopoulos DM. Generalized additive models for location, scale and shape, (with discussion). Appl. Statist., 54, part 3, pp 507–554. 2005.
- Wood SN. Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models. Journal of the Royal Statistical Society (B) 73(1):3–36.2011.
- Bates D, Maechler M, Bolker B, Walker S. Fitting Linear Mixed-Effects Models Using Ime4. Journal of Statistical Software, 67(1), 1–48.2015.
- 51. R Development Core Team. R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing; 2010.
- Eames KT, Tilston NL, White PJ, Adams E, Edmunds WJ. The impact of illness and the impact of school closure on social contact patterns. Health technology assessment. 2010; 14(34):267–312. https://doi.org/10.3310/hta14340-04 PMID: 20630125.
- Rocha LE, Masuda N. Individual-based approach to epidemic processes on arbitrary dynamic contact networks. Scientific reports. 2016; 6:31456. https://doi.org/10.1038/srep31456 PMID: 27562273
- Ten Bosch QA, Clapham HE, Lambrechts L, Duong V, Buchy P, Althouse BM, et al. Contributions from the silent majority dominate dengue virus transmission. PLoS Pathog. 2018; 14(5):e1006965. https:// doi.org/10.1371/journal.ppat.1006965 PMID: 29723307
- Vazquez-Prokopec GM, Perkins TA, Waller LA, Lloyd AL, Reiner RC Jr., Scott TW, et al. Coupled Heterogeneities and Their Impact on Parasite Transmission and Control. Trends in parasitology. 2016; 32 (5):356–67. https://doi.org/10.1016/j.pt.2016.01.001 PMID: 26850821
- Smith DL, Perkins TA, Reiner RC Jr., Barker CM, Niu T, Chaves LF, et al. Recasting the theory of mosquito-borne pathogen transmission dynamics and control. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2014; 108(4):185–97. https://doi.org/10.1093/trstmh/tru026 PMID: 24591453
- 57. Harrington LC, Fleisher A, Ruiz-Moreno D, Vermeylen F, Wa CV, Poulson RL, et al. Heterogeneous feeding patterns of the dengue vector, Aedes aegypti, on individual human hosts in rural Thailand. PLoS neglected tropical diseases. 2014; 8(8):e3048. https://doi.org/10.1371/journal.pntd.0003048 PMID: 25102306